Author's response to reviews

Title: A feasibility study examining the effect on lung cancer diagnosis of offering a chest x-ray to higher risk patients with chest symptoms: protocol for a randomized controlled trial

Authors:

Christopher N Hurt (hurtcn@cardiff.ac.uk)
Kirsty Roberts (RobertsKA3@cardiff.ac.uk)
Trevor K Rogers (Trevor.Rogers@dbh.nhs.uk)
Gareth O Griffiths (griffithsg@cardiff.ac.uk)
Kerry Hood (hoodk1@cardiff.ac.uk)
Hayley Prout (prouth@cardiff.ac.uk)
Annmarie Nelson (nelsona9@cardiff.ac.uk)
Jim Fitzgibbon (FitzgibbonJ1@cardiff.ac.uk)
Allan Barham (allan.barham@tiscali.co.uk)
Emma Thomas-Jones (thomas-jonese@cardiff.ac.uk)
Rhiannon Tudor-Edwards (r.t.edwards@bangor.ac.uk)
Seow T Yeow (s.t.yeo@bangor.ac.uk)
William Hamilton (W.Hamilton@exeter.ac.uk)
Angela Tod (A.Tod@shu.ac.uk)
Richard D Neal (r.neal@bangor.ac.uk)

Version: 3 Date: 8 November 2013

Author's response to reviews:

See covering letter